Author’s response to reviews

Title: Arterial Resection and Reconstruction in Pancreatectomy: Surgical Technique and Outcomes

Authors:
Qiyi Zhang (drqyzhang@163.com)
Jingjin Wu (8017048@zju.edu.cn)
Yang Tian (TianY@163.com)
Jixuan Duan (DuanJX@163.com)
Yi Shao (ShaoY@163.com)
Sheng Yan (YANS@zju.edu.cn)
Weilin Wang (wam@zju.edu.cn)

Version: 2 Date: 29 Jun 2019

Author’s response to reviews:

Dear Editor:

Thank you for your suggestion. We have carefully revised the manuscript according to the reviewers' comments, and also have re-scrutinized to improve the English by a native English speakers.

Reviewer reports:

1 Eirik Kjus Aahlin, M.D.:

We have changed the conclusions that the procedure is feasible.

2 Chang Moo Kang, Md, PhD

We have compared patients with pancreatic adenocarcinoma who received surgery with patients who received chemotherapy only (Table 4 and Figure 4). The oncologic outcome was unremarkable between two groups with the median survival 11.6 months in the operation group vs. 8.5 months in the chemotherapy group. p>0.05.